Author’s response to reviews

Title: Targeted next generation sequencing as a tool for precision medicine

Authors:

Markus Gulilat (bgulilat@uwo.ca)
Tyler Lamb (tyler.lamb@ucalgary.ca)
Wendy Teft (wendyteft@gmail.com)
Jian Wang (jwang@robarts.ca)
Jacqueline Dron (jdron@uwo.ca)
John Robinson (robinson@robarts.ca)
Rommel Tirona (rommel.tirona@schulich.uwo.ca)
Robert Hegele (hegele@robarts.ca)
Richard Kim (richard.kim@lhsc.on.ca)
Ute Schwarz (ute.schwarz@lhsc.on.ca)

Version: 1 Date: 30 Jan 2019

Author’s response to reviews:

We have revised our manuscript as requested by the Editorial Office including corrections in format, clarification of clip art design in Figure 1, and addition of information regarding informed consent of minors.